Author's response to reviews

Title: Study protocol: Determinants of participation and quality of life of adolescents with cerebral palsy: a longitudinal study (SPARCLE2)

Authors:

Allan F Colver (allan.colver@ncl.ac.uk)
Heather O Dickinson (heather.dickinson@ncl.ac.uk)
SPARCLE Group [SPARCLE] (allan.colver@ncl.ac.uk)

Version: 10 Date: 20 May 2010

Author's response to reviews:

Dear Sir
I had to make the following adjustments:
1 A sentence had been omitted at the end of the ethics section and I have reinserted this
I changed AC to AFC in referring to an author in the text
I reformatted the last bullet point under dissemination
Yours sincerely
Allan Colver